Author’s response to reviews

Title: A community-based intervention to improve dietary habit and to promote physical activity among the elderly: A cluster randomized trial

Authors:

Mika Kimura (mika-kimura@ilsijapan.org)
Ai Moriyasu (amoriyasu@ilsijapan.org)
Shu Kumagai (kumagai@human.ac.jp)
Taketo Furuna (furuna@sapmed.ac.jp)
Shigeko Akita (sakita@ilsijapan.org)
Shuichi Kimura (skimura@ilsijapan.org)
Takao Suzuki (suzutaka@ncgg.go.jp)

Version: 2 Date: 22 February 2012

Author’s response to reviews:

Dr. Rachel Neilan
Executive Editor
BMC Geriatrics
February 24, 2012

Dear Dr. Neilan,

Please find enclosed our revised manuscript entitled “A community-based intervention to improve dietary habit and to promote physical activity among the elderly: A cluster randomized trial”, which we would like to submit for publication as a research article in BMC Geriatrics.

In this revised manuscript, we added the clinical trial number at the last line of the abstract.

We are looking forward to hearing from you at your earliest convenience.

Yours sincerely,

Mika Kimura, MSc.
Project manager, Center for Health Promotion
International Life Sciences Institute Japan
Kojimachi RK Bld. 2-6-7 Kojimachi
Chiyoda-ku, Tokyo
102-0083 Japan
E-mail: mika-kimura@ilsijapan.org